Author's response to reviews

Title: What is Implementation Science and Why Should I Care? An Introduction for the Non-Specialist

Authors:

Mark S Bauer (mark.bauer@va.gov)
Laura Damschroder (laura.damschroder@va.gov)
Hildi Hagedorn (hildi.hagedorn@va.gov)
Jeffrey Smith (jeffrey.smith6@va.gov)
Amy M Kilbourne (amy.kilbourne@va.gov)

Version: 2
Date: 8 June 2015

Author's response to reviews:

Greetings Rannie,

Thank you for your review of our manuscript's format.

We didn’t comply with the formal Debate format initially because I understood from Tim and Anna that the manuscript didn’t fit squarely with any category but that Debate was how we should submit it.

It was not a problem to quickly convert the format, and also to adhere to the other guidelines:
• Author emails to cover page
• Abstract sections for Debate
• Text sections for Debate
• Table of abbreviations
• Line numbering
• Author contributions
• The research does not involve human subjects so I omitted that statement.

Don’t hesitate to let me know if I can provide further updates.

Thanks,
Mark
Cc: Anna/Tim

Mark S. Bauer, M.D.
Professor of Psychiatry, Harvard Medical School
Associate Director, Center for Healthcare Organization and Implementation Research (CHOIR)
VA Boston Healthcare System-152M
Jamaica Plain Campus, Building 9, 2nd floor
150 South Huntington Avenue
Boston, MA 02130
ph: 857-364-6380
fx: 857-364-6140
www.choir.research.va.gov